Author's response to reviews

Title: Low frequency of defective mismatch repair in a population-based series of upper urothelial carcinoma

Authors:

Kajsa M Ericson (Kajsa.Ericson@onk.lu.se)
Anna P Isinger (Anna.Isinger@onk.lu.se)
Bjorn L Isfoss (Bjorn.Logi.Isfoss@skane.se)
Mef Nilbert (Mef.Nilbert@onk.lu.se)

Version: 3 Date: 25 February 2005

Author's response to reviews: see over
Dear Editor,

Re: MS: 1321073595490987 "Low Frequency of Defective Mismatch Repair in a Population-based Series of Upper Urothelial Carcinoma".

Thank you for the rapid and valuable review of our manuscript.

According to your instructions, we have made the following revisions in the manuscript.

1) Title – All unnecessary capitals have been removed.
2) The keywords section has been removed.
3) The abstract is now structured into Background, Methods, Results, Conclusions.
4) In section headings unnecessary capitals are removed
5) The introduction is renamed to 'Background', Materials & Methods is renamed to 'Methods'.
6) The red highlighting is removed.
7) The figures are minimized
8) The tables are included at the end of the manuscript.
9) Small typing errors are corrected.

Sincerely,

Kajsa Ericson, MD, PhD

Department of Oncology
Lund University Hospital
SE-221-85 LUND, Sweden
Phone: +46-46-178525, Fax: +46-46-147327
E-mail: Kajsa.Ericson@onk.lu.se